Author’s response to reviews

Title: Assessment of risk factors for early childhood caries at different ages in Shandong, China and reflections on oral health education: a cross-sectional study

Authors:

Meng Zhang (zhangmeng1128@sdu.edu.cn)
Xinyue Zhang (1342980248@qq.com)
Yuan Zhang (ebmzhangyuan@yeah.net)
Yanan Li (1194917841@qq.com)
Chunchun Shao (819845062@qq.com)
Shijiang Xiong (xiongsj@sdu.edu.cn)
Jing Lan (lanjing@sdu.edu.cn)
Zhifeng Wang (kqwzf@sdu.edu.cn)

Version: 4 Date: 30 Mar 2020

Author’s response to reviews:

OHEA-D-19-00826
Assessment of risk factors for early childhood caries at different ages in Shandong, China and reflections on oral health education: a cross-sectional study
Meng Zhang; Xinyue Zhang; Yuan Zhang; Yanan Li; Chunchun Shao; Shijiang Xiong; Jing Lan; Zhifeng Wang.
BMC Oral Health

Dear Prof. Mohammad Reza Khami

Thank you very much your considering our manuscript for potential publication, and thank you very much for your comments. We have revised the manuscript, here is the point-by-point response letter. Thank you very much!

Editor Comments:

1. Please confirm whether informed consent obtained from all participants' parents/guardians was written or verbal, and clearly state this in your Ethics approval and consent to participate section. If verbal, please state the reason and whether the ethics committee approved this procedure.
All participants’ guardians approved and signed the informed consent, we’ve clearly states this in Ethics approval and consent to participate section (Declarations, lines 403-404, page 19).

2. Please remove the response to reviewers document from the end of your manuscript file as this is no longer needed in the publication process.
   All right. Thank you very much!

3. Please respond to the final comments from the Editor, found at the end of this email.
   Line 322: reference # 44 in the reference list is not Edward Lo.
   Thank you very much! We’ve revised it in line 322 (Discussion, page 15).

4. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

   Ok, all files have been uploaded as you requested. Thank you!